Author’s response to reviews

Title: Construction and Application of Service Quality Evaluation System in the Preclinical Research on Cardiovascular Implant Device

Authors:

yongchun Cui (cyc.fuwai.submission@gmail.com)
Fuliang Luo (fuliangluo@gmail.com)
Boqing Yang (boqingyang@126.com)
Bin Li (libin4936@163.com)
Qi Zhang (15810904688@163.com)
Gopika das (Gopika-ds@gmail.com)
Guangxin Yue (yueguangxin12345@sina.com)
Jiajie Li (jiajieruc@163.com)
Yue Tang (tangyue1226@vip.sina.com)
Xin Wang (fuwaiwangxin@188.com)

Version: 2 Date: 02 Feb 2019

Author’s response to reviews:

To: BMC medical informatics and decision making
Fu Wai Hospital, Chinese Academy of Medical Sciences and Peking Union Medical College, Beijing, 100037, China.
E-mail address: cyc.fuwai.submission@gmail.com
Yue Tang MD.

Dear Editor,

I am sending here with revised manuscript, which I would like to submit to BMC medical informatics and decision making. The paper is entitled: Construction and Application of Service Quality Evaluation System in the Preclinical Research on Cardiovascular Implant Devices (submission number MIDM-D-18-00398) by Yongchun Cui, Fuliang Luo, Boqing Yang, et al.
We will fully address reviewers’ comments point-by-point as followings.

1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).
   R: Thank you very much for your valuable suggestions. We have added the authors’ contribution in the Authors’ Contributions section and sent 'Request for change in authorship form' to the editorial office via email.

2. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section.
   R: added.
   YCC, YT, XW and JJL have contributed in study design. FLL, BQ Y and BL have contributed in acquisition and analysis of data. GXY has contributed in interpretation of data. YCC and QZ have contributed in drafting and development of the final manuscript. Gd has critically evaluated the draft article. All authors reviewed and approved final version of the manuscript.

3. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
   R: corrected

4. Please clarify one of the authors names as it differs in the manuscript and submission system. Should their name appear as Das Gopika' or 'Gopika Das'? Please amend accordingly.
   R: Corrected. Gopika Das is right.

5. We note that the current submission contains some textual overlap with other previously published works, in particular: https://bmcmedinformdecismak.biomedcentral.com/articles/10.1186/s12911-017-0409-5.
   This overlap mainly exists in the Results sections.
   While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrased these sections to minimise overlap.
   R: Yes, thank you very much. We have re-phrased these sections. Please see the contents between the 168th line and the 175th line of the text.

6. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If no acknowledgements are necessary then please state 'None'.
   R: Corrected.

7. Please assign to the section ' Standards, Technology, Machine Learning and Modelling'.
   R: Added

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
   R: Thanks, it has been revised.
I hope your favorable consideration for publication to BMC medical informatics and decision making. Thank you very much for your attention and consideration. I'm looking forward to hearing from you soon.

Sincerely

Yue Tang